Author's response to reviews

Title: The invasive lobular carcinoma as a prototype luminal A breast cancer: a retrospective cohort study

Authors:

So-Youn Jung (goje1@medimail.co.kr)
Junsoo Jeong (cbr400f@daum.net)
Seung-Ho Shin (sshhere@hanmail.net)
Youngmee Kwon (ymk@ncc.re.kr)
Eun-A Kim (euna25@hanmail.net)
Kyoung Lan Ko (ffrog92@naver.com)
Kyung Hwan Shin (radiat@ncc.re.kr)
Keun Seok Lee (kslee@ncc.re.kr)
In Hae Park (parkih@ncc.re.kr)
Seeyoun Lee (seeyoun@naver.com)
Seok Won Kim (surgeon69@medimail.co.kr)
Han-Sung Kang (rorerr@ncc.re.kr)
Jungsil Ro (jungsro@ncc.re.kr)

Version: 4 Date: 1 December 2010

Author’s response to reviews: see over
Dear Editor,

Thank you for your positive answer. We revised the following article as you recommended.
“The invasive lobular carcinoma as a prototype luminal A breast cancer: a retrospective cohort study”

We sincerely hope that the revised manuscript could be reviewed as acceptable for publication in *BMC cancer*.

Best regards,
Jungsil Ro, M.D.
Center for Breast Cancer,
National Cancer Center, Goyang, Korea.